# **JGR** Atmospheres

### **RESEARCH ARTICLE**

10.1029/2022JD037219

#### **Key Points:**

- Bulk transfer coefficients exhibit a substantial increase at low wind speed
- The increase is explained by wind gustiness and capillary wave roughness
- At higher wind speed, drag coefficient and Stanton number decrease with lake surface area

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

# Correspondence to:

S. Guseva, guseva@uni-landau.de

#### Citation:

Guseva, S., Armani, F., Desai, A. R., Dias, N. L., Friborg, T., Iwata, H., et al. (2023). Bulk transfer coefficients estimated from eddy-covariance measurements over lakes and reservoirs. *Journal of Geophysical Research: Atmospheres*, *128*, e2022JD037219. https://doi.org/10.1029/2022JD037219

Received 31 MAY 2022 Accepted 9 JAN 2023

#### **Author Contributions:**

Conceptualization: A. R. Desai, N. L. Dias, H. Iwata, T. Vesala, A. Lorke Data curation: S. Guseva, F. Armani, N. L. Dias, T. Friborg, H. Iwata, J. Jansen, G. Lükő, I. Repina, A. Rutgersson, T. Sachs, K. Scholz, U. Spank, P. Torma Formal analysis: S. Guseva, H. Iwata, A. Lorke

Funding acquisition: A. Lorke Investigation: S. Guseva, A. R. Desai, N. L. Dias, H. Iwata, J. Jansen, K. Scholz, U. Spank, V. Stepanenko, P. Torma, A. Lorke

Methodology: S. Guseva, A. R. Desai, N. L. Dias, H. Iwata, J. Jansen, I. Mammarella, K. Scholz, U. Spank, V. Stepanenko, T. Vesala, A. Lorke

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# **Bulk Transfer Coefficients Estimated From Eddy-Covariance Measurements Over Lakes and Reservoirs**

S. Guseva<sup>1</sup> , F. Armani<sup>2</sup>, A. R. Desai<sup>3</sup> , N. L. Dias<sup>4</sup> , T. Friborg<sup>5</sup>, H. Iwata<sup>6</sup> , J. Jansen<sup>7,8</sup> , G. Lükő<sup>9</sup> , I. Mammarella<sup>10</sup> , I. Repina<sup>11,12</sup>, A. Rutgersson<sup>13</sup> , T. Sachs<sup>14</sup>, K. Scholz<sup>15</sup>, U. Spank<sup>16</sup>, V. Stepanenko<sup>12,17,18</sup> , P. Torma<sup>9</sup>, T. Vesala<sup>10,19</sup> , and A. Lorke<sup>1</sup>

<sup>1</sup>Institute for Environmental Sciences, University of Koblenz-Landau, Landau, Germany, <sup>2</sup>Federal University of Paraná, Curitiba, Brazil, 3Department of Atmospheric and Oceanic Sciences, University of Wisconsin-Madison, Madison, WI, USA, <sup>4</sup>Department of Environmental Engineering, Federal University of Paraná, Curitiba, Brazil, <sup>5</sup>Department of Geosciences and Natural Resource Management, Copenhagen, Denmark, 6Department of Environmental Science, Faculty of Science, Shinshu University, Matsumoto, Japan, <sup>7</sup>Department of Ecology and Genetics/Limnology, Uppsala University, Uppsala, Sweden, <sup>8</sup>Département des Sciences Biologiques, Groupe de Recherche Interuniversitaire en Limnologie, Université du Québec à Montréal, Montréal, QC, Canada, 9Department of Hydraulic and Water Resources Engineering, Budapest University of Technology and Economics, Budapest, Hungary, <sup>10</sup>Institute for Atmospheric and Earth System Research/Physics, Faculty of Science, University of Helsinki, Helsinki, Finland, <sup>11</sup>A.M. Obukhov Institute of Atmospheric Physics, Moscow, Russia, <sup>12</sup>Research Computing Center, Lomonosov Moscow State University, Moscow, Russia, <sup>13</sup>Department of Earth Sciences, Uppsala University, Uppsala, Sweden, <sup>14</sup>GFZ German Research Centre for Geosciences, Potsdam, Germany, <sup>15</sup>Department of Ecology, University of Innsbruck, Innsbruck, Austria, <sup>16</sup>Technische Universität Dresden, Faculty of Environmental Sciences, Institute of Hydrology and Meteorology, Chair of Meteorology, Tharandt, Germany, 17Faculty of Geography, Lomonosov Moscow State University, Moscow, Russia, <sup>18</sup>Moscow Center of Fundamental and Applied Mathematics, Moscow, Russia, <sup>19</sup>Institute for Atmospheric and Earth System Research/Forest Sciences, Faculty of Agriculture and Forestry, University of Helsinki, Helsinki, Finland

**Abstract** The drag coefficient, Stanton number and Dalton number are of particular importance for estimating the surface turbulent fluxes of momentum, heat and water vapor using bulk parameterization. Although these bulk transfer coefficients have been extensively studied over the past several decades in marine and large-lake environments, there are no studies analyzing their variability for smaller lakes. Here, we evaluated these coefficients through directly measured surface fluxes using the eddy-covariance technique over more than 30 lakes and reservoirs of different sizes and depths. Our analysis showed that the transfer coefficients (adjusted to neutral atmospheric stability) were generally within the range reported in previous studies for large lakes and oceans. All transfer coefficients exhibit a substantial increase at low wind speeds ( $<3 \text{ m s}^{-1}$ ), which was found to be associated with the presence of gusts and capillary waves (except Dalton number). Stanton number was found to be on average a factor of 1.3 higher than Dalton number, likely affecting the Bowen ratio method. At high wind speeds, the transfer coefficients remained relatively constant at values of  $1.6 \cdot 10^{-3}$ ,  $1.0 \cdot 10^{-3}$ , respectively. We found that the variability of the transfer coefficients among the lakes could be associated with lake surface area. In flux parameterizations at lake surfaces, it is recommended to consider variations in the drag coefficient and Stanton number due to wind gustiness and capillary wave roughness while Dalton number could be considered as constant at all wind speeds.

**Plain Language Summary** In our study, we investigate the bulk transfer coefficients, which are of particular importance for estimation the turbulent fluxes of momentum, heat and water vapor in the atmospheric surface layer, above lakes and reservoirs. The incorrect representation of the surface fluxes above inland waters can potentially lead to errors in weather and climate prediction models. For the first time we made this synthesis using a compiled data set consisting of existing eddy-covariance flux measurements over 23 lakes and 8 reservoirs. Our results revealed substantial increase of the transfer coefficients at low wind speeds, which is often not taken into account in models. The observed increase in the drag coefficient (momentum transfer coefficient) and Stanton number (heat transfer coefficient) could be associated with the presence of wind gusts and capillary waves. In flux parameterizations at lake surface, it is recommended to consider them for accurate flux representation. Although the bulk transfer coefficients were relatively constant at high wind speeds, we found that the Stanton number systematically exceeds the Dalton number (water vapor transfer coefficient), despite the fact they are typically considered to be equal. This difference may affect the Bowen ratio method and result in biased estimates of lake evaporation.

#### 1 of 20





Resources: F. Armani, A. R. Desai, N. L. Dias, T. Friborg, H. Iwata, J. Jansen, G. Lükő, I. Mammarella, A. Rutgersson, T. Sachs, U. Spank, P. Torma, T. Vesala Software: S. Guseva, A. Lorke Supervision: A. Lorke Visualization: S. Guseva Writing – original draft: S. Guseva, A. Lorke

Writing – review & editing: S. Guseva, F. Armani, A. R. Desai, N. L. Dias, T. Friborg, H. Iwata, J. Jansen, G. Lükő, I. Mammarella, I. Repina, A. Rutgersson, T. Sachs, K. Scholz, U. Spank, V. Stepanenko, P. Torma, T. Vesala, A. Lorke

#### 1. Introduction

The major process that governs the interaction between the atmosphere and surface waters is the turbulent exchange of momentum, heat and gases at the air-water interface. Although lakes and reservoirs occupy only about 3% of the land surface area (Downing et al., 2006), they are known to have an impact on local weather and climate. For example, lakes affect the stability of the atmosphere above (Sun et al., 1997), leading to the formation of clouds and precipitation on the shores (Changnon & Jones, 1972; Eerola et al., 2014; Kato & Takahashi, 1981; Thiery et al., 2016). Furthermore, lakes and reservoirs are recognized as significant contributors to the global carbon cycle by emitting significant amounts of carbon dioxide and methane (DelSontro et al., 2018; Rosentreter et al., 2021).

The past three decades have seen a rapid development of lake models (Stepanenko et al., 2014) and their incorporation into numerical weather and climate prediction models (Ljungemyr et al., 1996; Mironov et al., 2010; Salgado & Le Mogne, 2010). Experiments on the coupling of lakes and the atmospheric model revealed their beneficial impact on the weather prediction quality (Balsamo et al., 2012). A number of case studies have demonstrated the importance of lakes for extreme local weather phenomena, such as lake-effect snow over Great American lakes (Fujisaki-Manome et al., 2020), deep hazardous convection over Great African lakes (Thiery et al., 2016), wind speeds over Lake Superior (Desai et al., 2009), or stratiform cloudiness in winter over Lake Ladoga (Eerola et al., 2014). Thus, an accurate representation of the exchange of momentum, heat and water vapor at the air-water interface in water bodies is essential to understand those processes.

In state-of-the-art, momentum, sensible and latent heat fluxes are usually determined based on gradient approaches utilizing transfer coefficients (also known as bulk transfer coefficients) and easy to measure meteorological and limnological variables, that is, wind speed, air temperature, air humidity and water surface temperature (Stull, 1988). The exchange at the air-water interface and therewith the bulk coefficients are controlled by boundary-layer turbulence. The bulk exchange coefficient of momentum, known as the drag coefficient  $(C_D, C_{DN})$  (Garratt, 1977), is of particular importance for all air-water fluxes. The coefficients of heat  $(C_H, C_{HN})$ and water vapor exchange  $(C_E, C_{EN})$  are also known as Stanton and Dalton numbers, respectively. Here, "N" stands for "neutral" transfer coefficients, corresponding to the neutral thermal stability of the atmosphere. The transfer coefficients depend on the measurement height of the mean wind speed, air temperature and humidity, respectively, and for this reason, they are usually reported for the reference meteorological height of 10 m.

A considerable amount of studies has been published on the momentum flux and the drag coefficient starting from the early 1950s when the fundamental work, presenting the theory later on named as Monin-Obukhov similarity theory, was published (Monin & Obukhov, 1954; Obukhov, 1971). The theory aims at describing the structure of turbulence in the atmospheric surface layer about several tens of meters thick with the assumption of the fluxes being constant and independent of height. Similarity laws introduce functional relations to derive the universal shapes for the vertical profiles of different quantities for atmospheric thermal stability other than neutral. During the past decades, considerable effort has been devoted to define the exact form of these similarity functions (Businger et al., 1971; Högström, 1988; Paulson, 1970; Zilitinkevich & Calanca, 2010).

As the drag coefficient is one of the key parameters in atmospheric and lake models, the errors in its parameterization lead to errors in the bulk flux estimates. Therefore, numerous early studies focused on exploring different parameterizations of the drag coefficient over the land and oceans in terms of wind speed, atmospheric stability, and surface roughness, which are a function of the surface wavefield (for oceans) (Garratt, 1977; Kantha & Clayson, 2000). Most of the extensive field measurement campaigns over the oceans have been conducted during the last 30 years of the twentieth century (Fairall et al., 1996; Godfrey & Beljaars, 1991; Large & Pond, 1981; Smith et al., 1996). Several of these studies agreed that the drag coefficient linearly increases with increasing wind speed ignoring the state of the wavefield. More recent parameterizations of the drag coefficient (e.g., the COARE algorithm by Fairall et al. (2003); Edson et al. (2013)), however, include a wave dependence. There is still an ongoing scientific discussion concerning the importance of waves and how their impact could be included in the models (Wu et al., 2019).

Along with the studies in the marine environment, the research started to focus on the drag coefficient estimated from measurements over large and medium-sized lakes (e.g., Donelan, 1982; Graf et al., 1984; Hicks, 1972; Simon, 1997). To date, in total, about two dozen studies focusing on lakes have been published since the beginning of the 1970s. In reviewing these studies, we separated them by the wind speed regime they were interested in.

It is usually assumed that surface wave development starts when the wind speed exceeds  $3-4 \text{ m s}^{-1}$  (Ataktürk & Katsaros, 1999; Kantha & Clayson, 2000). This is also supported by wave measurements in several lakes (Guseva et al., 2021; Simon, 1997). Therefore, we intend to separate the two wind speed regimes using this threshold.

At the "high" wind speed regime (wind speed exceeds  $3 \text{ m s}^{-1}$ ), in the most simplified way, the surface waves are assumed to be fully developed, and the surface roughness length is described as a function of wind stress, which is commonly known as Charnock relationship (Charnock, 1955). However, this assumption might not hold for lakes with limited wind fetch (Donelan, 1990; Geernaert, 1990). Thus, some research has been made to study the drag coefficient as a function of the surface wave state, for example, taking into account wave characteristics such as the wave age (Ataktürk & Katsaros, 1999; Donelan, 1982). Vickers and Mahrt (1997) reported that for a given wind speed the drag coefficient tends to be larger for younger steeper waves representative of short wind fetches than for longer fetches. Ataktürk and Katsaros (1999) could significantly reduce the scatter in the estimated drag coefficients by considering waves in the parameterization of the surface roughness length. However, these studies mainly examined large lakes and only a few were performed in lakes with short fetch and young wave states (Babanin & Makin, 2008; Lükő et al., 2020). Given the fact that the surface wave measurements in lakes are often not available, their effect still could be investigated via analyzing the relationship between the drag coefficient and fetch length.

At the "low" wind speed regime, several studies found that the neutral drag coefficient in lakes and oceans tended to increase by an approximate factor of two up to ten compared to the value of  $1.3 \cdot 10^{-3}$  (corresponding to a typical value of open water surface roughness, Foken, 2008) (Woolway et al., 2017; Wüest & Lorke, 2003). Although the wind speed dependence is obvious, many numerical and empirical studies employ a constant value for the drag coefficient, which is often considered as a model tuning parameter (Stepanenko et al., 2014). Despite the fact that there have been many attempts to address the reasons of such increase, there is still no consensus in the scientific community. The low wind speed regime was first described as the aerodynamically smooth flow, when the surface waves are buried within the viscous sublayer and the surface roughness is described as a function of the thickness of this layer (Schlichting, 1968). On the contrary, Wu (1988) proposed that the flow is aerodynamically rough and that capillary gravity waves play a key role at low wind speeds. Surface roughness length was described as a function of the water surface tension. As an additional reason for the increase of the drag coefficient at low wind speed, Godfrey and Beljaars (1991) and Grachev et al. (1998) considered the concept of gustiness, which assumes that at "zero" wind speeds there are dry random convective motions-gusts-in the convective boundary layer (CBL). Thus, the "traditional" formulation of the drag coefficient has been modified using the scalar-averaged wind speed (not the vector-averaged wind speed) to account for gusts. All the possible mechanisms mentioned above were addressed in the recent work by Wei et al. (2016). They concluded that none of them explained the increase of the drag coefficient at low wind speeds. However, they found it can be explained by the increase in the turbulent kinetic energy due to buoyant energy production. Similar to Grachev et al. (1998), Sahlée et al. (2014) and Liu et al. (2020) related the increase of the drag coefficient with nonlocal effects, such as the penetration of large convective eddies into the surface layer from the atmosphere above. Liu et al. (2020) introduced the factor describing this effect and estimated it from two-level measurements of wind speed (however, over the land surface and only for neutral conditions). Another formulation of the drag coefficient at low wind speeds was done by Zhu and Furst (2013) relating the drag coefficient to the turbulent kinetic energy budget. However, their fitting coefficients for the drag coefficient formula were found to be site-specific (Liu et al., 2020).

Other studies on the bulk transfer coefficients in lakes branched off from the main direction—potential physical mechanisms—with a focus on the possible correlation between the bulk transfer coefficients and some lake characteristics. Among them are lake depth at the measurement location (Panin et al., 2006), lake surface area (Read et al., 2012; Woolway et al., 2017), wind fetch at the measurement location (Lükő et al., 2020) and lake biota, for example, submerged macrophytes (Xiao et al., 2013). All studies showed a strong dependence of the transfer coefficients on these lake characteristics. The drag coefficient tends to decrease with increasing water depth, lake area, fetch and in the presence of water plants at the water surface. It is important to note that although Panin et al. (2006) and Woolway et al. (2017) revealed the correlation between the transfer coefficients and the lake parameters, the estimation of the transfer coefficients was based either on bulk parameterization (Woolway et al., 2017), or was compared to other studies where there were no direct flux measurements (Panin et al., 2006).

Fewer studies have been published on the Stanton and Dalton numbers. Although the measurements in the oceans showed their obvious increase at low wind speeds, both transfer coefficients were considered as fairly constant



with a value of  $1.1 \cdot 10^{-3}$  (review of these measurements in Kantha and Clayson (2000)). First measurements conducted in lakes revealed this value being higher and equal to ~ $1.5 \cdot 10^{-3}$  (Harbeck, 1962; Hicks, 1972) or  $1.9 \cdot 10^{-3}$  (Strub & Powell, 1987). Harbeck (1962) and Brutsaert and Yeh (1970) reported a dependence of the Dalton number on the lake surface area. Heikinheimo et al. (1999) summarized that the Dalton number is generally known to be less dependent on the wind speed. From the most recent studies (Dias & Vissotto, 2017; Li et al., 2016; Wei et al., 2016; Xiao et al., 2013), there is evidence that both coefficients depend on the wind speed and that the Stanton number is higher than the Dalton number by approximately a factor of 1.3. This indicates that the earlier assumption of the equality of both coefficients may not be valid for lakes.

The eddy-covariance (EC) technique is a micrometeorological method to directly measure momentum, heat, water vapor and greenhouse gas fluxes (Foken, 2008). It is based on the correlation between turbulent fluctuations of vertical wind speed and scalar air properties. Using this technique, one can obtain the spatial and temporal average of turbulent fluxes originating from an area called footprint and a period of meteorological stationarity (Burba & Anderson, 2010; Foken et al., 2012; Lenschow et al., 1994; Sun et al., 2006). Nowa-days, the EC technique is commonly used over lakes (Blanken et al., 2000; Golub et al., 2021; Lee et al., 2014; Mammarella et al., 2015; Nordbo et al., 2011; Spank et al., 2020; Vesala et al., 2006). However, several studies reported difficulties in measuring the wind stress at weak winds, which resulted in large uncertainties (Kantha & Clayson, 2000). Low wind speed conditions are more relevant for lakes and specifically small lakes that are the most abundant inland water bodies (Downing et al., 2006).

In this study, we evaluate the first multiple water body estimates of bulk transfer coefficients and their dependencies on wind speed and water body characteristics using EC data measured above lakes. The analysis aimed at answering the following research questions: (a) What are the typical values for the bulk transfer coefficients and their variability among lakes and reservoirs? (b) How do the values compare with the reported transfer coefficients for oceans and other lakes? (c) Can the mechanistic approaches mentioned above describe the transfer coefficients at low wind speed regime? (d) Is there a consistent dependence of the transfer coefficients on lake characteristics, such as water depth, lake area and wind fetch? In the sections below, we examine possible answers.

#### 2. Materials and Methods

#### 2.1. Eddy-Covariance Data Set

For this analysis, most of the existing EC data measured by various researchers over lakes and reservoirs were extracted from open access databases and repositories of published papers. The fluxes that are reported in the datasets were calculated using different software (e.g., EddyPro, LI-COR, Inc, 2021; TK3, Mauder & Foken, 2015; EddyUH, Mammarella et al., 2016). In total, we obtained data for 23 lakes and 8 reservoirs located in the arctic, subarctic, temperate and subtropical zones (Figure 1, Table S1 in Supporting Information S1). The water bodies are located in different landscapes, including mountains (e.g., Lake Lunz, Austria or Lake Klöntal, Switzerland), forests (e.g., Lake Vanajavesi, Finland), and arctic landscapes. The EC mast at each lake or reservoir was installed either on a floating or bottom-fixed platform, on shore, or on small islands. The measurement height ranged between 1.3 and 16.1 m with 2 m being the most frequent height among all data sets. Elongated shapes of the lakes or shore/island locations were the subject of wind direction filtering to ensure that the measured surface fluxes were originating from water. Approximately half of the water bodies in this study had a surface area (A  $[km^2]$ ) smaller than 10 km<sup>2</sup> with an average wind fetch ( $F_{ave}$  [m]) ranging from 160 to 1,550 m. The fetch grid was estimated from the map as the distance from the measurement location to the shore with the corresponding wind direction. Then, the time series of the fetch was interpolated from this grid using the measured wind directions. The average fetch was calculated as the mean distance for the filtered wind directions. The rest of the lakes and reservoirs were larger: the maximum surface area of  $2.6 \cdot 10^4$  km<sup>2</sup> and the maximum mean fetch of  $2.6 \cdot 10^4$  m refer to one of the North-American Great Lakes—Lake Erie in the USA. The maximum depth  $(D_{max} [m])$  varied between 1.3 m (Lake Villasjön, Sweden) and 89 m (Rappbode Reservoir, Germany). Each EC data set contained the estimated variables averaged over 30 min intervals.

The variables included wind speed ( $U_z$  [m s<sup>-1</sup>]), wind direction (WD [°]), friction velocity ( $u_*$  [m s<sup>-1</sup>]) as a quantity characterizing the momentum flux ( $\tau = \rho_a u_*^2$  [kg m<sup>-1</sup> s<sup>-2</sup>],  $\rho_a$ -air density [kg m<sup>-3</sup>]), air temperature ( $T_a$  [°C]), turbulent fluxes of sensible heat (H [W m<sup>-2</sup>]), and latent heat ( $L_v E$  [W m<sup>-2</sup>]), the latter referred to in this paper as





**Figure 1.** (a) Geographical distribution of the eddy-covariance measurements over the lakes and reservoirs used in this study (red circles). Map was created using the software by Pawlowicz (2020). (b) The maximum depth versus surface area for each lake or reservoir. The circles in blue and red color show lakes with average wind fetch less than or more than 1,000 m, respectively, with the lake name next to it. The diameter of the circle represents the relative average fetch: the larger diameter, the larger the average fetch is.

water vapor heat flux as well. Water temperature was provided either as skin surface temperature ( $T_s$  [°C]) or bulk water temperature, measured at 0–0.5 m water depth ( $T_w$  [°C]). The skin temperature was observed with an infrared thermometer or calculated from outgoing longwave radiation, both corrected for the reflectance of incoming longwave radiation. Some lakes or reservoirs had only momentum flux data, resulting in fewer estimates of heat and water vapor transfer coefficients. Precipitation (parameter which was considered as a factor for filtering the data) was not available for all data sets. The duration of the EC measurements ranged from 11 days (Lake Wohlen, Switzerland) to 2,243 days (or ~6.1 years, Lake Dagow, Germany) with a median duration of 155 days.

#### 2.2. Data Filtering and Averaging

The individual data sets used in the analysis were subject to filtering with the following different criteria.

- 1. filtering based on stationarity and integral turbulence test quality flags;
- 2. restriction of the wind directions to ensure >90% of footprint was originated from water;
- 3. removing periods with ice cover;
- 4. removing periods with precipitation (if data on precipitation was available);
- 5. removing periods when the difference between water surface temperature (surface water specific humidity) and air temperature (air specific humidity) is less than  $0.2^{\circ}$ C ( $1.5 \cdot 10^{-3}$  kg kg<sup>-1</sup>);
- 6. removing periods with floating vegetation on the water surface (only for Lake Suwa, Japan).

Quality screening of EC data is known to be site- and instrument-specific (Burba & Anderson, 2010). The data were either available in filtered form, or they contained the quality flags provided by the software. Non-filtered data sets included quality flags for each flux value (momentum, sensible and latent heat fluxes) to ensure the stationarity of the time series (homogeneity of the flow) and developed turbulent conditions (Foken et al., 2004; Foken & Wichura, 1996).

Removing wind directions was site-specific and we carefully studied each individual site. We only accepted the data from periods when wind was blowing from the lake with sufficient fetch. We specified the accepted wind directions for each site in Table S1 in Supporting Information S1. We focused on open-water conditions and discarded ice-covered periods either using the water temperature time series or interval camera data. For Lake Suwa we removed the approximate periods when floating vegetation appeared on the water surface using interval camera data, however, for other sites this kind of data was not available. For some sites, all erroneous data due to rain interference and site maintenance were filtered by data providers, or we removed periods with precipitation (if data were available). Based on the standard accuracy of temperature and humidity sensors, we



also applied a minimum threshold for temperature and specific humidity difference between water surface and air. We describe the effect of these filters on the data in Text S1 in Supporting Information S1. Note, that in SI and in the sections below all transfer coefficients are shown as median values following suggestions from former studies (DeCosmo et al., 1996; Fairall et al., 2003).

#### 2.3. Transfer Coefficients

Turbulent fluxes of momentum ( $\tau$ ), sensible heat (H) and latent heat ( $L_{\nu}E$ ) at the water surface are expressed as:

$$\tau = \rho_a \left[ \left( \overline{u'w'} \right)^2 + \left( \overline{v'w'} \right)^2 \right]^{1/2} = \rho_a u_*^2 = \rho_a C_D U_{10}^2, \tag{1a}$$

$$H = \rho_a c_p \overline{w'T'} = -\rho_a c_p T_* u_* = \rho_a c_p C_H U_{10} (T_s - T_{10}), \tag{1b}$$

$$L_v E = \rho_a L_v \overline{w'q'} = -\rho_a L_v q_* u_* = \rho_a L_v C_E U_{10}(q_s - q_{10}), \tag{1c}$$

where u', v', w' are the horizontal and vertical wind velocity fluctuations, respectively;  $\overline{w'T'}$  [m s<sup>-1</sup> K],  $\overline{w'q'}$  [m s<sup>-1</sup> kg kg<sup>-1</sup>] are the covariances of vertical wind velocity and air temperature (T') and specific humidity (q') fluctuations.  $U_{10}$  [m s<sup>-1</sup>] is wind speed at 10 m height,  $T_s$  and  $T_{10}$  [K] are the surface water temperature and the air temperature at 10 m height, respectively,  $q_s$  and  $q_{10}$  [kg kg<sup>-1</sup>] are the specific humidity at the air-water interface (estimated from surface temperature) and at 10 m height, respectively.  $c_p$  [J kg<sup>-1</sup> K<sup>-1</sup>] is the specific heat of air

at constant pressure, and  $L_v$  [J kg<sup>-1</sup>] is the latent heat of vapourization.  $T_* = \frac{-\overline{w'T'}}{u_*}$  and  $q_* = \frac{-\overline{w'q'}}{u_*}$  are temperature and specific humidity scales, respectively. The standard sign convention is that the momentum flux is defined as positive downward, while sensible and latent heat fluxes as positive upward (Kaimal & Finnigan, 1994). Equations 1a–1c are not the only way to describe the transfer coefficients. The widely used COARE 3.0 bulk algorithm (Fairall et al., 2003) includes scalar-averaged wind speed instead of  $U_{10}$  (vector-averaged wind speed) in Equations 1a–1c, which includes the gustiness (will be discussed below in Section 2.4.4).

Using measured flux data from the obtained EC datasets, the transfer coefficients can be derived from Equations 1a-1c as follows:

$$C_D = \frac{u_*^2}{U_{10}^2},\tag{2a}$$

$$C_H = \frac{\overline{w'T'}}{U_{10}(T_{\rm s} - T_{10})},\tag{2b}$$

$$C_E = \frac{w'q'}{U_{10}(q_{\rm s} - q_{10})}.$$
 (2c)

Wind speed, air temperature  $(T_z)$  and specific humidity  $(q_z)$  measured at a certain height z were converted to a standard height of 10 m considering stability of the atmosphere following the equations:

$$U_{10} = U_z - \frac{u_*}{\kappa} \left[ \ln\left(\frac{z}{10}\right) - \psi_u\left(\frac{z}{L}\right) + \psi_u\left(\frac{10}{L}\right) \right],\tag{3a}$$

$$T_{10} = T_z - \frac{T_*}{\kappa} \left[ \ln\left(\frac{z}{10}\right) - \psi_T\left(\frac{z}{L}\right) + \psi_T\left(\frac{10}{L}\right) \right],\tag{3b}$$

$$q_{10} = q_z - \frac{q_*}{\kappa} \left[ \ln\left(\frac{z}{10}\right) - \psi_T\left(\frac{z}{L}\right) + \psi_T\left(\frac{10}{L}\right) \right],\tag{3c}$$

where  $\kappa$  is the von Kármán constant, L [m] is the Obukhov length,  $\psi_u\left(\frac{z}{L}\right)$  is the stability function which is the integral of the empirical universal function for the momentum flux and  $\psi_T\left(\frac{z}{L}\right)$ —the same for sensible and latent heat (Businger et al., 1971). In the literature, z/L is usually denoted as the non-dimensional stability parameter  $\zeta$ . To remove the effect of atmospheric stability on the magnitude of the transfer coefficients,  $C_D$ ,  $C_H$ ,  $C_E$  are



converted to their neutral counterparts  $C_{\text{DN}}$ ,  $C_{\text{HN}}$ ,  $C_{\text{EN}}$  (i.e., for neutrally-stratified atmospheric conditions) (Large & Pond, 1981):

$$C_{\rm DN} = \kappa^2 \left[ \ln \left( \frac{10}{z_0} \right) \right]^{-2} = C_D \left[ 1 + \kappa^{-1} C_D^{\frac{1}{2}} \psi_u \left( \frac{10}{L} \right) \right]^{-2}, \tag{4a}$$

$$C_{\rm HN} = C_D \left[ 1 + \kappa^{-1} C_D^{\frac{1}{2}} \psi_u \left( \frac{10}{L} \right) \right]^{-1} \left[ \frac{C_D}{C_H} + \kappa^{-1} C_D^{\frac{1}{2}} \psi_T \left( \frac{10}{L} \right) \right]^{-1}, \tag{4b}$$

$$C_{\rm EN} = C_D \left[ 1 + \kappa^{-1} C_D^{\frac{1}{2}} \psi_u \left( \frac{10}{L} \right) \right]^{-1} \left[ \frac{C_D}{C_E} + \kappa^{-1} C_D^{\frac{1}{2}} \psi_T \left( \frac{10}{L} \right) \right]^{-1}, \tag{4c}$$

where  $z_0$  is the surface roughness length. For our calculations, we used the Kansas-type stability functions (Businger et al., 1971) in the form of Högström (1988), which is the most frequently applied form (Foken, 2008).  $C_{\text{DN}}$ ,  $C_{\text{HN}}$ ,  $C_{\text{EN}}$  were estimated for 31, 25, 23 water bodies under study, respectively, depending on the flux data availability (see details about each lake or reservoir in Table S1 in Supporting Information S1 and in data repository).

In the scientific community, there has been an ongoing discussion on the form of the transfer coefficients to be presented. For example, some studies focused only on neutral values of the drag coefficient (Li et al., 2016) or some considered the drag coefficient non-adjusted to their neutral counterpart ( $C_D$ ). Other studies addressed the so-called "effective" drag coefficient, which was derived as the slope coefficient for the linear relationship between  $u_{*}^2$  and  $U_{10}^2$  (Xiao et al., 2013). We examine the difference between  $C_D$ ,  $C_H$ ,  $C_E$  and  $C_{DN}$ ,  $C_{HN}$ ,  $C_{EN}$  in Section 3.1.

#### 2.4. Parametrizations of the Drag Coefficient at Low and High Wind Speeds

#### 2.4.1. Smooth Flow

Previous studies focused on the parameterizations of surface roughness length  $z_0$  (see Equation 4a) to assess wind speed dependence of the drag coefficient (e.g., Ataktürk & Katsaros, 1999). In our study, we compared  $C_{DN}$ estimated from measured momentum fluxes with the existing approaches. One of the approaches is based on the smooth flow regime at low wind speed (<3 m s<sup>-1</sup>), where the thickness of the viscous sublayer ( $\delta_v$ ) determines the aerodynamic roughness of the interface (Schlichting, 1968), and not the physical roughness of the water surface:

$$\delta_{\nu} = z_0 = \alpha \frac{\nu}{u_*},\tag{5}$$

where  $\alpha = 0.11$  [-] and  $\nu = 1.6 \times 10^{-5}$  [m<sup>2</sup> s<sup>-1</sup>] is kinematic viscosity of air.  $z_0$  can be derived from Equation 4a as:

$$z_0 = z \exp\left(-\frac{\kappa}{\sqrt{C_{\rm DN}}}\right). \tag{6}$$

#### 2.4.2. Capillary Waves

As an alternative method to parameterize  $C_{\text{DN}}$  at low wind speeds, we considered the approach proposed by Wu (1994). He suggested that the wind shear stress in the absence of large gravity waves is related to the ripples (capillary waves). For the capillary waves, the roughness length is related to surface tension ( $\sigma$ ) as:

$$z_0 = \alpha_{\rm Wu} \frac{\sigma}{\rho_w u_*^2},\tag{7}$$

where  $\alpha_{Wu} = 0.18$  [-] is an empirical constant and  $\rho_w$  [kg m<sup>-3</sup>] is water density. Surface tension at a temperature of 20°C is  $\sigma = 7.28 \cdot 10^{-2}$  N m<sup>-1</sup>.

#### 2.4.3. Charnock Relationship

With increasing wind speed, the thickness of the viscous sublayer becomes smaller, and the aerodynamic roughness of the water surface  $(z_0)$  becomes minimal, before surface gravity waves evolve. At wind speeds exceeding



3 m s<sup>-1</sup>, waves protrude from the viscous sublayer and surface roughness length increases with increasing wind speed, indicating the transition from a smooth to a rough flow regime. Charnock (1955) proposed the following equation for surface roughness length over fully developed surface waves, which account for typical oceanic conditions:

$$z_0 = \gamma \frac{u_*^2}{g},\tag{8}$$

where  $\gamma$  ranges from 0.011 to 0.0185 [-] (Garratt, 1994), g [m s<sup>-2</sup>] is the gravitational acceleration.

#### 2.4.4. The Concept of Gustiness

Under strong convective conditions, the wind stress at the water surface is governed by random convective motions—gusts—in the convective boundary layer (CBL), whereas the mean wind speed vector can even become zero (Godfrey & Beljaars, 1991). These large convective eddies embrace the entire CBL and affect the turbulence regime in the atmospheric surface layer. Grachev et al. (1998) formulated an approach to estimate the drag coefficient using this concept. In their study, the gustiness could explain the apparent increase of the drag coefficient estimated using the traditional equation (Equations 2a, 4a) at low wind speeds. The estimated drag coefficient accounting for gusts was a factor of 1.5–6 smaller at wind speeds below 2 m s<sup>-1</sup> in comparison to the drag coefficient calculated from Equations 2a, 4a. The gustiness concept is widely accepted and used in the COARE algorithm to estimate air-sea fluxes (Fairall et al., 2003).

The gustiness factor, G corresponds to the ratio of the scalar-averaged ( $\tilde{U}_{10}$ , for the definition see (Grachev et al., 1998) to vector-averaged wind speed:

$$G_{\rm wind} = \frac{\tilde{U}_{10}}{U_{10}}.$$
 (9)

Taking into account that  $\tilde{U}_{10}^2 = U_{10}^2 + U_g^2$ , where  $U_g \cong \sigma_u^2 + \sigma_v^2$  is a gustiness velocity ( $\sigma_u$ ,  $\sigma_v$  are standard deviations of the wind speed components), the gustiness factor can be estimated as

$$G_{\text{wind}}^2 = \frac{U_{10}^2 + U_g^2}{U_{10}^2} = \frac{U_{10}^2 + \left(\sigma_U^2 + \sigma_V^2\right)}{U_{10}^2} = 1 + \frac{\left(\sigma_U^2 + \sigma_V^2\right)}{U_{10}^2}.$$
 (10)

Since the scalar-averaged wind speed is not a standard output parameter in the EC-software, the estimation of *G* from velocity standard deviations allowed calculation of the transfer coefficients for more lake data sets. Alternatively, *G* can be parameterized in terms of the convective velocity scale  $w_*$  (Fairall et al., 2003; Grachev et al., 1998). We denote the gustiness factors derived from measured wind speed and from CBL scaling as  $G_{wind}$  and  $G_{conv}$ , respectively:

$$G_{\rm conv}^2 = 1 + \left(\frac{\beta w_*}{U_{10}}\right)^2,\tag{11}$$

where  $\beta = 1.2$  [-] is an empirical constant (Beljaars, 1995) and  $w_*$  is expressed as:

$$w_* = \left(g z_{\text{CBL}} \frac{\overline{w' T_v'}}{T_v}\right)^{1/3},\tag{12}$$

where,  $T_v$  [K] is the virtual temperature,  $z_{CBL}$  [m] is the CBL height, defined as the height of the lowest inversion. Previous studies used the fixed height of the CBL equal to 600 m (Fairall et al., 2003) or 1,000 m (Beljaars, 1995). The neutral gustiness drag coefficient  $C_{DNG}$  and its gustiness counterpart  $\tilde{C}_{DG}$  is:

$$C_{\rm DNG} = \left[ C_{\rm DG}^{-1/2} G + \kappa^{-1} \psi_u \left( \frac{10}{L} \right) \right]^{-2}, \tag{13a}$$

and other gustiness transfer coefficients, C<sub>HNG</sub> and C<sub>ENG</sub>, can be derived as:

$$C_{\rm HNG} = \left[ C_{\rm DG}^{-1/2} G + \kappa^{-1} \psi_u \left( \frac{10}{L} \right) \right]^{-1} \left[ C_{\rm DG}^{1/2} C_{\rm HG}^{-1} + \kappa^{-1} \psi_T \left( \frac{10}{L} \right) \right]^{-1}, \tag{13b}$$



$$C_{\rm ENG} = \left[ C_{\rm DG}^{-1/2} G + \kappa^{-1} \psi_u \left( \frac{10}{L} \right) \right]^{-1} \left[ C_{\rm DG}^{1/2} C_{\rm EG}^{-1} + \kappa^{-1} \psi_T \left( \frac{10}{L} \right) \right]^{-1},$$
(13c)

where  $C_{DG} = \left(\frac{\tilde{u}_*}{U_{10}}\right)^2$  and  $\tilde{u}_*$  is the scalar-averaged friction velocity. The majority of the available EC datasets from lakes did not contain the scalar-averaged friction velocity. Akylas et al. (2003) investigated the combinations with different averaging procedures and suggested that vector-averaged friction velocity  $u_*$  is more appropriate to use with scalar-averaged wind speed for all wind speed classes. Thus, we used the vector-averaged friction velocity and therefore  $C_{DG} = C_D$ ,  $C_{HG} = C_H$ ,  $C_{EG} = C_E$ .

In a first step, we applied the gustiness approach for the drag coefficient with the cases corresponding to unstable atmospheric condition ( $\zeta < 0$ ). We estimated two types of the gustiness drag coefficient, using as  $G_{\text{wind}}$  and  $G_{\text{conv}}$  and we considered the former one as a reference. These calculations were possible only for a subset of the data sets (11 lakes) for which both scalar-averaged wind speed (or the standard deviations of the wind speed components) and virtual sensible heat flux  $(\overline{w'T'_v})$  were available. Then we fitted the coefficient  $\beta$  in Equation 11 in order to get best agreement between the parametrized  $C_{\text{DNG}}$  and the referenced one. As the second step, we applied the parametrization of gustiness with the fitted  $\beta$  (Equation 11) to larger subset of the data (26 lakes) for which the virtual sensible heat flux was available. Thus, we estimated all gustiness transfer coefficients (Equations 13a–13c).

We compared the gustiness transfer coefficients with the standard formulations (Equations 4a-4c) and with the results from the COARE 3.0 bulk algorithm (Fairall et al., 2003), which includes the parameterization of gustiness (Equations 11 and 12). In addition, the COARE algorithm considers the effect of changing water surface roughness by using parameterizations of  $z_0$  that combines the smooth flow approach (Equation 5) and Charnock's relationship (Equation 8).

#### 3. Results

#### 3.1. Transfer Coefficients Over Lakes

Bulk transfer coefficients for neutral atmospheric stability  $C_{DN}$ ,  $C_{HN}$  and  $C_{EN}$  (Equations 4a–4c) were estimated using data from 23 lakes and 8 reservoirs (see data availability details in Table in the data repository). The transfer coefficients varied between the water bodies and differed on average by a factor of 2–3 for wind speeds exceeding 3 m s<sup>-1</sup>. However, we identified three water bodies for which the estimated drag coefficients ( $C_{DN}$ ) were exceptionally large at all wind speeds (up to a factor of five, Lake Quinghai, China, Nam Theun 2 Reservoir, Laos), or exceptionally low (factor of four, Bol'shoi Vilyui Lake, Russia), when compared to other water bodies with similar surface area. These three water bodies contributed largely to the variability among systems (Figure S5a in Supporting Information S1 shows the estimates for individual water bodies). We did not find possible sources of errors and considered these data as outliers. In the overall estimates and in the range of variability shown in Figure 2a, we included the complete data set.

All transfer coefficients showed a similar wind speed dependence (Figures 2a–2c). At high wind speeds (>3 m s<sup>-1</sup>),  $C_{\text{DN}}$ ,  $C_{\text{HN}}$ ,  $C_{\text{EN}}$  had relatively constant values of  $1.8 \cdot 10^{-3}$ ,  $1.4 \cdot 10^{-3}$ ,  $10 \cdot 10^{-3}$ , respectively. All transfer coefficients increased towards the lowest wind speeds. The strongest increase was found for  $C_{\text{DN}}$ , which was one order of magnitude higher ( $1.1 \cdot 10^{-2}$ ) at the lowest wind speed ( $0.5 \text{ m s}^{-1}$ , the first bin) compared to values at higher wind speeds. A similar, but less pronounced increase was observed for  $C_{\text{HN}}$  and  $C_{\text{EN}}$ : their values at the lowest wind speed were  $3.1 \cdot 10^{-3}$  and  $2.2 \cdot 10^{-3}$ , respectively.

Unstable atmospheric conditions ( $\zeta < 0$ ) prevailed over all water bodies, particular during the evening and at night time, when >80% of all data were obtained under unstable conditions (Figure 2d). Stable atmospheric conditions occurred most frequent during the day (12–19 hr). In addition, we estimated the percentage of time when the wind speed was less than 3 m s<sup>-1</sup> (Figure 2e). Low wind speed conditions prevailed slightly during the evening and at night, when the atmosphere was mostly unstable. This means that the significant increase of the transfer coefficients at low wind speeds frequently coincides with unstable atmospheric conditions, when the water is still warm and the atmosphere starts cooling at the end of the day.

To analyze the effect of atmospheric stability on the transfer coefficients, we compared the transfer coefficients  $(C_D; C_H, C_E, \text{Equations 2a-2c})$  with their neutral counterparts  $(C_{DN}; C_{HN}, C_{EN}, \text{Equations 4a-4c}, \text{Figure S6 in Supporting Information S1})$ . We found that atmospheric stability did not significantly affect the values of  $C_D, C_H$  and  $C_E$  at wind speeds exceeding 3 m s<sup>-1</sup>: their values were in close agreement with  $C_{DN}, C_{HN}$  and  $C_{EN}$ . However,





**Figure 2.** Neutral (a) drag coefficient ( $C_{DN}$ ), (b) Stanton number (heat transfer coefficient,  $C_{HN}$ ), (c) Dalton number (water vapor transfer coefficient,  $C_{EN}$ ) versus wind speed at 10 m height. The shaded gray area indicates the variability among water bodies by marking the range from the 5th to the 95th percentiles of the median values estimated for each wind speed bin ( $0.5 \text{ m s}^{-1}$ ). The black line with circles, marked as L&R denoting "lakes and reservoirs," represents the median values for all lakes and reservoirs. The small inset in (a) shows the data beyond the scale. Vertical and horizontal black dashed lines mark a constant wind speed of 3 m s<sup>-1</sup> and typical values of  $C_{DN}$ ,  $C_{HN} = C_{EN} 1.3 \cdot 10^{-3}$ ,  $1.1 \cdot 10^{-3}$ , respectively. Colored lines show the results from previous studies: LK, brown line—Lake Kasumigaura, Japan (eddy covariance, Wei et al., 2016); LN, red line—Lake Neuchâtel, Switzerland (dissipation method, Simon, 1997); LG, pink line—nearshore site at Lake Geneva, Switzerland (wind profile method, Graf et al., 1984); RG, light orange color—Reservoir Gorkiy, Russia (wind profile method, Kuznetsova et al., 2016); OO, dark green color—open ocean (eddy covariance, Large & Pond, 1981); OO, light green color—open ocean (eddy covariance, Fairall et al., 2003); CO, dark yellow color—coastal ocean at limited fetch conditions (eddy covariance, Lin et al., 2002). (d) Mean diel pattern of the percentage of time periods with unstable atmospheric conditions (stability parameter  $\zeta < 0$ ). (e) Mean diel pattern of the percentage of time with low wind speed (<3 m s<sup>-1</sup>). The red and black lines in (d) and (e) show the mean value and the shaded area shows  $\pm$  standard deviation of all data.

it is evident that at low wind speeds (0–2 m s<sup>-1</sup>) these transfer coefficients under in-situ conditions were systematically higher (up to a factor of 2–3) than their neutral counterparts  $C_{\text{DN}}$ ,  $C_{\text{HN}}$  and  $C_{\text{EN}}$ .

Estimation of  $C_H$  and  $C_E$  (Equations 2b, 2c and Equations 4b, 4c) involves water surface temperature, for which the skin temperature is the most appropriate measure. However, these measurements were not available for some sites. Instead, we used water temperature measured at some depth (often varying between 0 and 0.5 m between datasets). We compared three types of calculations of  $C_H$  using two subsets which use: (a) only skin temperature (b) only water temperature and (c) the combined lake data set which includes either skin or water temperature (Figure S7 in Supporting Information S1).  $C_{\text{HN}}$  estimated with water temperature tends to be slightly lower than the estimates using skin temperature (the mean percentage difference is approximately 15%). As a result, we presented  $C_H$  and  $C_E$  (Figures 2b and 2c) calculated using all available data, independent of how water surface temperature was measured. When both the skin and water temperatures were available for one site, the skin temperature was used.

#### 3.2. Parametrizations of the Drag Coefficient

As described in Section 2.4.4, we estimated drag coefficients accounting for gustiness ( $C_{DNG}$ ) using gustiness factors derived from measured scalar-averaged wind speed ( $G_{wind}$ ), and from the parametrization of convective velocities ( $G_{conv}$ ). To test the applicability of the parametrization, we compared  $C_{DNG}$  estimated using both



# Journal of Geophysical Research: Atmospheres



**Figure 3.** Neutral gustiness transfer coefficients (a)  $C_{\text{DNG}}$ , (b)  $C_{\text{HNG}}$  and (c)  $C_{\text{ENG}}$  (calculated using  $G_{\text{conv}}$ , see Section 2.4.4) versus  $U_{10}$  (red line) estimated for 26, 23 and 23 lakes, respectively. The red shaded area marks the range between the 5th and 95th percentiles. The black line with symbols shows estimates of  $C_{\text{DN}}$ ,  $C_{\text{HN}}$  and  $C_{\text{EN}}$  from Figure 2. The dark yellow line in all plots shows the function  $C_{\text{DN}} = b_1 [1 + b_2 \exp(b_3 U_{10})]$  proposed by Liu et al. (2020) with fitted coefficients (Table 1). The green and blue lines show transfer coefficients as they are used in the COARE 3.0 algorithm for bulk parameterizations. The green line shows the original COARE parameterization, which combines the effects of thesmooth flow approach (Equation 5) and Charnock relationship (Equation 8). The blue line shows a modified parameterization, which includescapillary wave roughness (Equation 7 with  $\alpha_{\text{Wu}} = 0.8$ ) and Charnock relationship.

approaches during unstable atmospheric conditions for a subset containing 11 lake data sets with all required data (Figure S8 in Supporting Information S1). The estimated  $C_{\text{DNG}}$  based on  $G_{\text{conv}}$  was slightly higher than the one based on  $G_{\text{wind}}$ . We obtained a coefficient  $\beta = 1.4$  (compared to a value of 1.2 suggested by Beljaars (1995)) for the best agreement between both drag coefficients (minimum of the mean absolute value of their difference). Note, that for this subset of lake data set the gustiness drag coefficient was on a factor of two lower than  $C_{\text{DN}}$  (Figure S8 in Supporting Information S1).

Using the fitted parameterization, we calculated  $C_{\text{DNG}}$ ,  $C_{\text{HNG}}$  and  $C_{\text{ENG}}$  based on  $G_{\text{conv}}$  for a larger subset of the data—26 and 23 lakes (Figure 3).  $C_{\text{DNG}}$  and  $C_{\text{ENG}}$  decreased approximately by a factor of two at wind speeds less than 1.5 m s<sup>-1</sup> in comparison to the standard formulation  $C_{\text{DN}}$  and  $C_{\text{EN}}$ .  $C_{\text{ENG}}$  was almost constant with a value of  $1.0 \cdot 10^{-3}$  throughout all ranges of wind speed.  $C_{\text{HNG}}$  decreases only slightly (~10%) in comparison with  $C_{\text{HN}}$  at low wind speeds. For wind speeds exceeding 3 m s<sup>-1</sup>, all gustiness transfer coefficients almost coincided with their standard formulation, only  $C_{\text{DNG}}$  was slightly smaller: ~1.6  $\cdot 10^{-3}$ .

We examined the possible mechanisms (Equations 5 and 7, Section 2.4) that could explain the increase of the drag coefficient at low wind speed and the Charnock relationship (Equation 8), which describes its wind speed dependence at high wind speed. We applied the COARE 3.0 algorithm to the same subset of the data (26 and 23 lakes) and compared the results with our estimates of the gustiness transfer coefficients. It was evident that the smooth flow approach (together with Charnock relationship) could not explain the increase of  $C_{\text{DNG}}$  at low wind speeds.

#### Table 1

Coefficients for the Empirical Function  $C = b_1 [1 + b_2 \exp(b_3 U_{10})]$  (Liu et al., 2020), Describing the Wind Speed Dependence of the Bulk Transfer Coefficients of Momentum ( $C_{\text{DN}}$ ), Heat ( $C_{\text{HN}}$ ), and Water Vapor ( $C_{\text{EN}}$ )

	$b_1$	$b_2$	$b_3$
$C_{ m DN}$	$1.7 \cdot 10^{-3}$	1	-1.1
$C_{ m HN}$	1.3.10-3	1.5	-0.8
$C_{ m EN}$	$1.1 \cdot 10^{-3}$	1	-1

*Note.* The coefficients were obtained from least-square fits of the Function to the bin-averaged data shown in Figures 3a–3c.

However, replacing this model with the approach which considers capillary waves (Equation 7 with  $\alpha_{Wu} = 0.8$ ) led to reasonable agreement between the bulk parameterization and  $C_{DNG}$  estimated from measured fluxes. In contrast,  $C_{ENG}$  calculated using this approach overestimated values of the EC-derived  $C_{ENG}$ . There, the smooth flow was a more appropriate parameterization.

When analyzing the standard formulation of the drag coefficient, we found that the function describing the wind speed dependence of  $C_{\rm DN}$  proposed by Liu et al. (2020) based on EC measurements over terrestrial surfaces  $(C_{\rm DN} = b_1 [1 + b_2 \exp(b_3 U_{10})])$ , can also be applied to observations over all lakes (Figure 3a). In a similar way, we applied this empirically derived function to  $C_{HN}$  and  $C_{EN}$  estimates (Figure S7). The fitted coefficients for our data are provided in Table 1.





**Figure 4.** Ratio of  $C_{\text{ENG}}$  to  $C_{\text{ENG}}$  estimated for each individual data set (21 water bodies, shown by gray lines). Black line with circles shows the median values for all data sets. The horizontal dashed black line shows a ratio of 1:1 and the horizontal dashed red line indicates the overall mean value of the ratio (1.3).

The gustiness approach, together with increasing surface roughness due to capillary waves allowed to explain the increase of the transfer coefficients at low wind speeds, suggesting that these formulations of the transfer coefficients can provide most accurate flux parameterizations at lake surfaces.

The mean ratio of  $C_{\rm HNG}$  to  $C_{\rm ENG}$  is 1.3 and has its maximum value of 2 at low wind speeds (Figure 4). For wind speeds ranging from 2 to 9 m s<sup>-1</sup> the ratio remained at relatively constant value of 1.2.

# **3.3.** Dependence of the Bulk Transfer Coefficients on the Lake Characteristics

We examined the dependencies of the bulk transfer coefficients accounting for gustiness on lake characteristics, including the maximum and average water depth, water depth at the measurement site, maximum and average wind fetch, and water surface area. As the transfer coefficients at high wind speeds were relatively constant, we first analyzed effects of lake characteristics on the median values of the transfer coefficients for wind speeds exceeding 3 m s<sup>-1</sup> estimated for each individual water body.

We found that  $C_{\text{DNG}}$  and  $C_{\text{HNG}}$  decreased significantly (Pearson correlation coefficient r = -0.5, *p*-value <0.05) with increasing lake surface area (Figures 5a and 5d). These relationships could be expressed as power law dependencies ( $y = x^A \exp(B \ln 10)$ ), where *A* and *B* are the slope and intercept

of the linear regression  $\log_{10} y = A \log_{10} x + B$  with exponent of -0.06. Most variability in  $C_{\text{DNG}}$  was found to be explained by the lake surface area (for log-transformed data the coefficient of determination was  $R^2 = 0.3$ ). The correlation between  $C_{\text{DNG}}$ ,  $C_{\text{HNG}}$  and  $C_{\text{ENG}}$  and mean or maximum fetch was low ( $r \sim -0.2$ , -0.3, Figures 5b, 5e and 5h, Figures S9a, S9d, S9g in Supporting Information S1). A principal component analysis revealed that lake surface area has a largest predictive power (Figure S10 in Supporting Information S1). We did not find a significant correlation ( $r \sim -0.3$ , p-value >0.05) between  $C_{\text{ENG}}$  and surface area, however, a similar trend as for  $C_{\text{DNG}}$  and  $C_{\text{HNG}}$  could be observed.

Using the principal component analysis, we identified that there was no significant correlation of the transfer coefficients at high wind speeds with maximum, average or local water depth (Figures 5c, 5f and 5i, Figures S9 and S10 in Supporting Information S1). We used the exponential dependence from Panin et al. (2006) to compare with our results. However, we did not have sufficient sites with larger depth to confirm the depth dependence reported in their study.

At low wind speeds (<3 m s<sup>-1</sup>), the transfer coefficients were strongly wind speed dependent (Figures 2a–2c) and their relationships with lake characteristics are examined separately for each different wind speed interval. Here we found that  $C_{\text{DNG}}$  significantly increased with increasing water surface area for wind speeds between 0.5 m s<sup>-1</sup> and 2 m s<sup>-1</sup>. At higher wind speeds these correlations become negative, as in the analysis for wind speed >3 m s<sup>-1</sup> presented above. As an example, we show the transfer coefficients for a wind speed of 1 m s<sup>-1</sup> in Figure S11 in Supporting Information S1. At the same time, we found significant correlation of  $C_{\text{DNG}}$  with measurement height at low wind speeds, which was not present at high wind speeds (Figure S11d in Supporting Information S1). No correlation with measurement height was found for  $C_{\text{HNG}}$  and  $C_{\text{ENG}}$ .

Additionally, we looked for a possible relationship between the averaged wind speed (estimated over entire time series for each individual water body) and surface area. We found a significant correlation between them in a double-logarithmic domain (r = 0.5, p-value <0.05, Figure S12 in Supporting Information S1), resulting in increasing mean wind speed with increasing lake size following a power-law dependence  $U_{10} = A_s^{0.05} \exp(0.5 \ln 10)$ .

#### 4. Discussion

#### 4.1. Bulk Transfer Coefficients Estimated for Lakes and Reservoirs

We examined the bulk transfer coefficients describing the transport of momentum, heat and water vapor at the water surface estimated based on EC data collected at 23 lakes and 8 reservoirs of different size, depth, and location. At first, we used their standard formulations commonly used in literature (Equations 4a-4c). All





**Figure 5.** Neutral transfer coefficients accounting for gustiness (a, b, c)  $C_{\text{DNG}}$ ; (d, e, f)  $C_{\text{HNG}}$ ; (g, h, i)  $C_{\text{ENG}}$  versus surface area of the water body, mean fetch length, and water depth at the measurement site. Panels show the median values of the transfer coefficients for wind speeds exceeding 3 m s<sup>-1</sup> for each individual water body. The red lines show the linear regressions of log-transformed transfer coefficients. The corresponding power laws as well as the Pearson correlation coefficient (*r*) and *p*-value are provided in the upper left corner of each panel. Significant correlation is marked by bold labels. The single red symbol in (a) marks the data from Nam Theun 2 Reservoir, which was not considered in the linear regression analysis and in the Pearson correlation as an obvious outlier. Blue, green and dark yellow lines show results from previous studies by Woolway et al. (2017), Panin et al. (2006), Harbeck (1962), respectively.

transfer coefficients tended to increase toward low wind speeds and remained relatively constant at wind speeds exceeding 3 m s<sup>-1</sup>. This increase was reported in previous studies for lakes (see, e.g., Wei et al., 2016; Xiao et al., 2013) and for the land surface (Grachev et al., 2011; Liu et al., 2020) and has been extensively investigated, yet remained unexplained up to now. Authors of Grachev et al. (2011) referred to lakes as an "intermediate"



case between over-sea and over-land locations in terms of turbulent transfer. The lower bound for  $C_{DN}$ ,  $C_{HN}$ ,  $C_{EN}$  among the water bodies at high wind speeds were within the range reported by previous studies, including for large lakes (>200 km<sup>2</sup> (Kuznetsova et al., 2016; Wei et al., 2016), classical open ocean measurements (Fairall et al., 2003; Large & Pond, 1981) and coastal sites under fetch-limited conditions (Lin et al., 2002). Indeed, we also considered large lakes (Figure 1b) that were expected to have the smallest drag coefficient as they had the largest fetch (e.g., Lake Erie, Lake Taihu, Lake Balaton). The mean  $C_{DN}$  for winds exceeding 3 m s<sup>-1</sup> was equal to  $1.8 \cdot 10^{-3}$  and this value corresponded to an upper bound for the water surface roughness (0.001 m) reported by Foken (2008), but was a factor of two higher than the values reported for oceans and large lakes or reservoirs (Fairall et al., 2003; Large & Pond, 1981).

Values of  $C_{DN}$  varied considerably depending on the type of measurements used for its estimation. For example, in Simon (1997),  $C_{DN}$  was calculated from the dissipation rate of turbulent kinetic energy measured at the water side of the air-water interface in the relatively large Lake Neuchâtel (218 km<sup>2</sup>, Switzerland).  $C_{DN}$  was significantly lower than our estimates (factor of ten) and the estimates from lakes or marine measurements (factor of five). However, these estimates also confirmed the increase of  $C_{DN}$  at low wind speeds.  $C_{DN}$  at high wind speeds calculated from the wind profile method at the nearshore site in Lake Geneva (Graf et al., 1984) was in close agreement with our estimates. The strong increase of  $C_{DN}$ ,  $C_{HN}$ ,  $C_{EN}$  at low wind speeds was similar to the one observed for a large lake using surface fluxes measured by EC method (Wei et al., 2016), but it was not supported by measurements in the marine environment.

#### 4.2. Bulk Transfer Coefficients at Low Wind Speed

Low wind speeds are typical conditions for lakes (Woolway et al., 2018), especially for smaller ones (Figure S10 in Supporting Information S1), which are most abundant by number (Downing et al., 2006). The most pronounced increase in bulk transfer coefficients at low wind speed was observed for  $C_{\rm DN}$ , which was up to one order of magnitude higher at low wind speeds compared to its value at high wind speeds. We found less pronounced increases of  $C_{\rm EN}$  and  $C_{\rm EN}$ , yet their values at low wind speed can be larger compared to the constant values at high wind speed by up to factor of three and two, respectively. Periods with low wind speeds mostly corresponded to periods with unstable atmospheric conditions or enhanced convective transport, which is the most prevailing condition at all studied lakes during the ice-free period. It confirms former findings of Read et al. (2012) and Woolway et al. (2017).

While Wei et al. (2016) suggested that the contribution of gusts (different formulation of the  $C_{\rm DN}$ ) was not significant in their measurements obtained over a large lake, we found that the strong increase can partially be attributed to missing consideration of gustiness in the definition of  $C_{\rm DN}$ . Correction involving the gustiness factor ( $G_{\rm conv}$ ) could reduce the values of  $C_{\rm DN}$  and  $C_{\rm EN}$  by up to a factor of two at wind speeds around 0.5 m s<sup>-1</sup>. Thus, the gustiness approach is recommended to be applied in bulk parametrizations of both fluxes in lakes.

It is important to note, that the CBL scaling requires knowledge about the height of the unstable boundary layer, for which we used a fixed value of 600 m, as in other studies, including the COARE algorithm for bulk parametrization of air-sea fluxes (Fairall et al., 2003). Over lakes, convective motions can be associated with lake-land breeze circulation (Crosman & Horel, 2012), which additionally depends on lake size and on the atmospheric conditions in the surrounding landscape. Nevertheless, we found that the gustiness parameterization based on the convective velocity scale ( $G_{conv}$ ) agreed reasonably well with observed gustiness in wind speed fluctuations ( $G_{wind}$ ). Best agreement was obtained by using a slightly higher value of the empirical constant  $\beta$  (Equation 11), for which we estimated a value of 1.4 (compared  $\beta = 1.2$  used in the COARE algorithm. The close agreement between parameterized and measured guestiness suggests that the parameterization can improve bulk parametrizations also under more complex CBL dynamics in lakes in comparison to the open ocean.

Comparison of our measurements with the bulk parametrizations in the commonly used COARE 3.0 algorithm, which includes the gustiness correction (Fairall et al., 2003), showed a remaining underestimation of the parameterized  $C_{\rm DNG}$  at low wind speed. Obviously, the remaining increase in the gustiness-corrected drag coefficient at low wind could not be described by the smooth flow parametrization (Equation 5), which is applied in the COARE 3.0. Instead, we applied the capillary wave approach for parametrizing the surface roughness at low wind speed (Equation 7) and found reasonable agreement between observed and parameterized gustiness drag coefficients. For  $C_{\rm ENG}$  the smooth flow model performed better, as it became a nearly constant value after applying the

gustiness approach. The effect of gustiness was not significant for  $C_{\text{HNG}}$  and similar to  $C_{\text{DNG}}$  the capillary wave parametrization performed better than smooth flow approach.

The wind speed dependence of the bulk transfer coefficients in their standard formulation (especially at low winds), could be well described by an empirical function that was originally proposed for the land surface (Liu et al., 2020). This suggests that this function can be used to describe the transfer coefficients without consideration of gustiness.

Unexpectedly, we found that the gustiness drag coefficient significantly increase with increasing lake surface areaat wind speeds less than 2 m s<sup>-1</sup>. This result is counterintuitive, because at low winds we did not expect a dependence on lake surface area or fetch, as it should only be important for the development of surface waves, which appears only at wind speeds exceeding 3 m s<sup>-1</sup> (Guseva et al., 2021; Simon, 1997). At the same time, we found a significant positive correlation between  $C_{\text{DNG}}$  and the measurement height, which was also unexpected. This finding could be the result of measurement limitations and require a separate detailed investigation.

#### 4.3. Bulk Transfer Coefficients at High Wind Speed

The estimated  $C_{\text{DN}}$  and  $C_{\text{DNG}}$  agreed closely at high wind speeds, thus, can be considered identically. In comparison to the results from the COARE 3.0 algorithm,  $C_{\text{DNG}}$  at wind speeds larger than 3 m s<sup>-1</sup> was higher than predicted by Charnock relationship (Equation 8). This result was expected as Charnock relationship is based on the assumption that the water surface roughness is controlled by fully developed surface gravity waves. When comparing with the results from the modified COARE 3.0 algorithm (capillary wave instead of smooth flow approach), we found that the consideration of Charnock relationship had a weak influence only, so that it could probably be omitted without large changes in bulk parameterizations.

We found a significant correlation between  $C_{\text{DNG}}$ ,  $C_{\text{HNG}}$  and lake surface area. For large lakes, the transfer coefficients at high wind speed tended to be lower. At these higher wind speeds, the surface gravity waves could potentially reach the fully developed state in large water bodies. For  $C_{\text{ENG}}$ , the correlation with lake surface area was not significant, but a similar trend could be observed. The values of  $C_{\text{ENG}}$  in our analysis were approximately a factor of two lower than in previous studies (Brutsaert & Yeh, 1970; Harbeck, 1962). Our results could not confirm a bilinear decrease of  $C_{\text{DNG}}$  with increasing lake size with a weaker dependence for large lakes, as suggested by Woolway et al. (2017) (Figure 5a). This can potentially be attributed to the parameterization of the transfer coefficients used in Woolway et al. (2017) in lack of flux measurements. In contrast to the results reported in (Panin et al., 2006), we did not find evidence for the existence of an influence of water depth on the bulk transfer coefficients.

#### 4.4. Study Limitations

The estimated bulk transfer coefficients show large scatter, even after filtering the data. The scatter is particularly high at light winds, that is, in the first three to four wind speed intervals  $(0.5-2 \text{ m s}^{-1})$ . It could be associated with limitations of the EC measurements, namely, the validity of the underlying assumptions, including the homogeneity and stationarity of the flow, as well as by increasing random errors. These effects may not have been removed completely by data filtering. Moreover, our estimates of  $C_{\text{DNG}}$  differ from former results obtained using different types of measurements, such as water-side energy dissipation rates (Figures 3a, e g., Simon, 1997). Thus, the combination of water- and air side measurements could be beneficial for further investigation of the bulk transfer coefficients.

The parameterization of gustiness used in the COARE 3.0 algorithm assumes the CBL height as a constant value of 600 m, however, this model has been used mostly for open ocean conditions, not for lakes. Considering the fact, that the thermal internal boundary layer can develop above lakes due to temperature difference between land and water (Glazunov & Stepanenko, 2015), the exact layer structure of the atmosphere (corresponding to the cases of the high drag coefficients—low wind speed and convection) should be studied.

In our analysis we considered 30-min intervals for averaging, however, the results will likely depend on it. For example, selection of longer time interval leads to larger gustiness velocity. But since the 30-min time interval is the standard averaging interval for the EC data processing, we did not have an opportunity to test the effect of averaging time scale.



Hwang (2004) suggested that the standard height of 10 m at which the transfer coefficients are reported is inappropriate for analyzing  $C_{\text{DN}}$  and its dependence on surface roughness under wave conditions. They argue that the only relevant parameter that could serve as a reference height is the wavelength that describes the decay rate of the waves with the distance from the water surface. The adjustment of the transfer coefficients to 10 m height may not be very relevant for lakes and reservoirs and the flux measurements at two different heights should be considered in future measurements. These measurements would additionally provide confirmation for the existence of a constant flux layer, which is another important assumption underlying EC measurements.

#### 4.5. Broader Implications

Bulk transfer coefficients are usually applied in numerical models for the atmospheric boundary layer, as well as in hydrodynamic models of lakes and reservoirs. Currently, the global modeling studies focusing on the lake mixing and phytoplankton blooms for climate change predictions use constant values for  $C_{\rm DN}$  (Grant et al., 2021; Jöhnk et al., 2008; Read et al., 2014; Woolway & Merchant, 2019), or consider  $C_{\rm DN}$  as a tuning parameter of the models (Stepanenko et al., 2014). Inadequate values of  $C_{\rm DN}$  result in biased estimates of the current velocities in lake models (Chen et al., 2020). The increase of the transfer coefficients at low wind speeds observed in our analysis can therefore lead to significant errors, as these conditions are the most prevailing conditions for lakes. We found that instead of the standard formulation of the transfer coefficients, their gustiness counterparts  $C_{\rm DNG}$ ,  $C_{\rm HNG}$ , and  $C_{\rm ENG}$  can improve flux parameterizations in lake models.

In the absence of data for the gustiness approach, the empirical parameterizations of the wind-speed dependence of the bulk transfer coefficients provided in Table 1 can potentially be applied in modeling lake-atmosphere interactions. The observed dependence on the lake surface area is more complicated to implement, as we observed contrary dependencies at low and high wind speeds.

While the Stanton and Dalton numbers are commonly assumed to be equal, we found that  $C_{\rm HNG}$  was on average by a factor of 1.3 higher than  $C_{\rm ENG}$  (averaged over all wind speeds and all water bodies under study). The finding of  $C_{\rm HNG}$  being higher than  $C_{\rm HNG}$  confirmed the results reported by, for example, (Dias & Vissotto, 2017; Wei et al., 2016). The mean value of  $C_{\rm HNG}$  for high wind speeds  $(1.0 \cdot 10^{-3})$  was found to be the same as in (Kantha & Clayson, 2000), but  $C_{\rm HNG}$  was larger  $(1.4 \cdot 10^{-3})$  as in (Harbeck, 1962; Hicks, 1972). The fact that  $C_{\rm HN} > C_{\rm EN}$  may have significant implications, because it results in biased estimates of lake evaporation based on the energy-budget (Bowen ratio method). In particular, it would result in larger (smaller) sensible heat (latent heat) fluxes than those predicted under that assumption. Both the physical mechanisms underlying their difference and the extent of the differences in the predicted sensible and latent heat fluxes require further investigation.

In state-of-the-art weather and earth system models, lakes are included as separate tiles in the model cells, where the surface fluxes over the tiles are computed via Monin-Obukhov similarity scaling. The models provide constant meteorological variables for each grid cell, which is a so-called blending height concept (von Salzen et al., 1996). To use the bulk transfer coefficients derived in this study to compute fluxes, specific values of wind, temperature and humidity over lakes should be used, which can be obtained in generalization of the tile approach, involving the parameterization of internal boundary layers over contrasting surfaces (Arola, 1999; de Vrese et al., 2016; Molod et al., 2003). MacKay (2019) presents a specific example of such an approach developed for lakes and wind speed only.

#### 5. Conclusions

We were the first to analyze the bulk transfer coefficients of momentum, sensible and latent heat from directly measured surface fluxes above various lakes and reservoirs. We observed a pronounced increase of the transfer coefficients at low wind speeds ( $<3 \text{ m s}^{-1}$ ) and relatively constant values at high wind speeds ( $>3 \text{ m s}^{-1}$ ). The strong increase in the transfer coefficients at low wind speed was found to be associated with the existence of gusts. It is recommended to use the gustiness approach in calculation of the transfer coefficients. The reduced, yet still increasing values of the gustiness drag coefficient and Stanton number at low wind speeds can be explained by capillary wave roughness. The gustiness Dalton number, however, remains constant at all wind speeds and is better described by the smooth flow roughness. The Stanton number was systematically higher than the Dalton number by a factor of 1.3, which has implications for the Bowen ratio method. At high wind speed, the drag coefficient and the Stanton number decreased with increasing surface area of the water body and with



increasing fetch length, whereas the opposite was found at low wind speed. No significant correlation was found between the transfer coefficients and lake depth. In a simplified approach, the bulk transfer coefficients can be calculated without consideration of gustiness by using empirical function that has been proposed for the land surface and that we fitted to data measured over lake and reservoir surfaces. The COARE algorithm adequately describes the gustiness drag coefficient and Stanton number if capillary wave roughness is considered instead of that for smooth flow. We underline the need for simultaneous measurements of waterside and airside turbulent fluxes in future investigations, as well as experimental confirmation of the validity of the assumptions underlying eddy-covariance flux measurements at low wind speed.

### **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

#### **Data Availability Statement**

Derived quantities such as the bulk transfer coefficients are available in the open data repository Zenodo: https://doi.org/10.5281/zenodo.6597828 (Guseva et al., 2022). Original datasets are available from the open data repositories or from the co-authors (see details in Table S1 in Supporting Information S1).

#### References

- Akylas, E., Tsakos, Y., Tombrou, M., & Lalas, D. P. (2003). Considerations on minimum friction velocity. Quarterly Journal of the Royal Meteorological Society, 129(591), 1929–1943. https://doi.org/10.1256/qj.01.73
- Arola, A. (1999). Parameterization of turbulent and mesoscale fluxes for heterogeneous surfaces. Journal of the Atmospheric Sciences, 56(4), 584–598. https://doi.org/10.1175/1520-0469(1999)056<0584:POTAMF>2.0.CO;2
- Ataktürk, S. S., & Katsaros, K. B. (1999). Wind stress and surface waves observed on Lake Washington. Journal of Physical Oceanography, 29(4), 633–650. https://doi.org/10.1175/1520-0485(1999)029<0633:WSASWO>2.0.CO;2
- Babanin, A. V., & Makin, V. K. (2008). Effects of wind trend and gustiness on the sea drag: Lake George study. Journal of Geophysical Research, 113(C2), C02015. https://doi.org/10.1029/2007JC004233
- Balsamo, G., Salgado, R., Dutra, E., Boussetta, S., Stockdale, T., & Potes, M. (2012). On the contribution of lakes in predicting near-surface temperature in a global weather forecasting model. *Tellus A: Dynamic Meteorology and Oceanography*, 64(1), 15829. https://doi.org/10.3402/ tellusa.v64i0.15829
- Beljaars, A. C. M. (1995). The parametrization of surface fluxes in large-scale models under free convection. Quarterly Journal of the Royal Meteorological Society, 121(522), 255–270. https://doi.org/10.1002/qj.49712152203
- Blanken, P. D., Rouse, W. R., Culf, A. D., Spence, C., Boudreau, L. D., Jasper, J. N., et al. (2000). Eddy covariance measurements of evaporation from Great Slave Lake, Northwest Territories, Canada. Water Resources Research, 36(4), 1069–1077. https://doi.org/10.1029/1999WR900338
- Brutsaert, W., & Yeh, G.-T. (1970). Implications of a type of empirical evaporation formula for lakes and pans. *Water Resources Research*, 6(4), 1202–1208. https://doi.org/10.1029/WR006i004p01202
- Burba, G. G., & Anderson, D. J. (2010). A brief practical guide to eddy covariance flux measurements: Principles and workflow examples for scientific and industrial applications. LI-COR Biosciences.
- Businger, J. A., Wyngaard, J. C., Izumi, Y., & Bradley, E. F. (1971). Flux-profile relationships in the atmospheric surface layer. Journal of the Atmospheric Sciences, 28(2), 181–189. https://doi.org/10.1175/1520-0469(1971)028<0181:FPRITA>2.0.CO;2
- Changnon, S. A., & Jones, D. M. A. (1972). Review of the influences of the Great Lakes on weather. *Water Resources Research*, 8(2), 360–371. https://doi.org/10.1029/WR008i002p00360
- Charnock, H. (1955). Wind stress on a water surface. Quarterly Journal of the Royal Meteorological Society, 81(350), 639–640. https://doi.org/10.1002/qi.49708135027
- Chen, F., Zhang, C., Brett, M. T., & Nielsen, J. M. (2020). The importance of the wind-drag coefficient parameterization for hydrodynamic modeling of a large shallow lake. *Ecological Informatics*, 59, 101106. https://doi.org/10.1016/j.ecoinf.2020.101106
- Crosman, E. T., & Horel, J. D. (2012). Idealized large-eddy simulations of sea and lake breezes: Sensitivity to lake diameter, heat flux and stability. *Boundary-Layer Meteorology*, 144(3), 309–328. https://doi.org/10.1007/s10546-012-9721-x
- De Cosmo, J., Katsaros, K. B., Smith, S. D., Anderson, R. J., Oost, W. A., Bumke, K., & Chadwick, H. (1996). Air-sea exchange of water vapor and sensible heat: The Humidity Exchange Over the Sea (HEXOS) results. *Journal of Geophysical Research*, 101(C5), 12001–12016. https:// doi.org/10.1029/95JC03796
- DelSontro, T., Beaulieu, J. J., & Downing, J. A. (2018). Greenhouse gas emissions from lakes and impoundments: Upscaling in the face of global change. *Limnology and Oceanography Letters*, 3(3), 64–75. https://doi.org/10.1002/lol2.10073
- Desai, A. R., Austin, J. A., Bennington, V., & McKinley, G. A. (2009). Stronger winds over a large lake in response to weakening air-to-lake temperature gradient. *Nature Geoscience*, 2(12), 855–858. https://doi.org/10.1038/ngeo693
- de Vrese, P., Schulz, J.-P., & Hagemann, S. (2016). On the representation of heterogeneity in land-surface–atmosphere coupling. *Boundary-Layer Meteorology*, *160*(1), 157–183. https://doi.org/10.1007/s10546-016-0133-1
- Dias, N. L., & Vissotto, D. (2017). The effect of temperature-humidity similarity on Bowen ratios, dimensionless standard deviations, and mass transfer coefficients over a lake: Temperature-humidity similarity over a lake. *Hydrological Processes*, *31*(2), 256–269. https://doi.org/10.1002/hyp.10925
- Donelan, A. (1990). Air-sea interaction. In The Sea (Vol. 9, pp. 239-292). Environment and Climate Change Canada.
- Donelan, M. (1982). The dependence of the aerodynamic drag coefficient on wave parameters. In Proceedings of First International Conference on Meteorology and Air-Sea Interaction of the Coastal Zone (pp. 381–387).

# Acknowledgments

S. Guseva and A. Lorke were financially supported by the German Research Foundation (Deutsche Forschungsgemeinschaft, DFG, grant LO1150/12-1). N. Dias and F. Armani were supported by research project FUNPAR 2882, with funding provided by CHESF (São Francisco Hydroelectric Company). A.R. Desai acknowledges support from the DOE Ameriflux Network Management Project and the NSF North Temperate Lakes LTER (#DEB-2025982). Funding for the AmeriFlux data portal was provided by the U.S. Department of Energy Office of Science. H. Iwata was supported by the Japan Society for the Promotion of Science KAKENHI Grants 17H05039 and 21H02315. J. Jansen was supported by the Swedish Research Council under VR Grant 2020-06460. G. Lükő was supported by the ÚNKP-21-3 New National Excellence Program of the Ministry for Innovation and Technology, Hungary. Data processing for Bol'shoi Vilvui Lake made by I. Repina was partially supported by a grant of the Russian Science Foundation 21-17-00249. T. Sachs was funded by a Helmholtz Young Investigators Grant (VH-NG-821) and infrastructure at Lake Dagow was operated as part of the Terrestrial Environmental Observatories Network (TERENO) of the Helmholtz Association of German Research Centers. K. Scholz was supported by the Austrian Academy of Sciences (ÖAW) as part of the project "Influence of climate extremes on C cycling dynamics across the boundaries of aquatic ecosystems (EXCARB)" and by the Autonome Provinz Bozen-Südtirol (ALCH4 Project) - both grants received by Prof. Georg Wohlfahrt (University of Innsbruck). The measurements in Bautzen Reservoir and U. Spank were supported by the German Science Foundation in frame of the projects TREGATA (project number 288267759) and MEDIWA (project number 445326344). V. Stepanenko was supported by Russian Ministry of Science and Higher Education (agreement No. 075-15-2019-1621). P. Torma was supported by the by the National Research. Development and Innovation Office (Grants K134559 and K138176). I. Mammarella and T. Vesala acknowledge funding from ICOS-Finland (University of Helsinki). We thank Christian Wille, Sarah Waldo and Werner Eugster for providing the data. Open Access funding enabled and organized by Projekt DEAL.

- Downing, J. A., Prairie, Y. T., Cole, J. J., Duarte, C. M., Tranvik, L. J., Striegl, R. G., et al. (2006). The global abundance and size distribution of lakes, ponds, and impoundments. *Limnology & Oceanography*, 51(5), 2388–2397. https://doi.org/10.4319/lo.2006.51.5.2388
- Edson, J. B., Jampana, V., Weller, R. A., Bigorre, S. P., Plueddemann, A. J., Fairall, C. W., et al. (2013). On the exchange of momentum over the open ocean. *Journal of Physical Oceanography*, 43(8), 1589–1610. https://doi.org/10.1175/JPO-D-12-0173.1
- Eerola, K., Rontu, L., Kourzeneva, E., Pour, H. K., & Duguay, C. (2014). Impact of partly ice-free Lake Ladoga on temperature and cloudiness in an anticyclonic winter situation – A case study using a limited area model. *Tellus A: Dynamic Meteorology and Oceanography*, 66(1), 23929. https://doi.org/10.3402/tellusa.v66.23929
- Fairall, C. W., Bradley, E. F., Hare, J. E., Grachev, A. A., & Edson, J. B. (2003). Bulk parameterization of air–sea fluxes: Updates and verification for the COARE algorithm. *Journal of Climate*, 16(4), 571–591. https://doi.org/10.1175/1520-0442(2003)016<0571:BPOASF>2.0.CO;2
- Fairall, C. W., Bradley, E. F., Rogers, D. P., Edson, J. B., & Young, G. S. (1996). Bulk parameterization of air-sea fluxes for Tropical Ocean-Global atmosphere Coupled-Ocean Atmosphere response experiment. *Journal of Geophysical Research*, 101(C2), 3747–3764. https:// doi.org/10.1029/95JC03205
- Foken, T. (2008). Micrometeorology. Springer Berlin Heidelberg. https://doi.org/10.1007/978-3-540-74666-9
- Foken, T., Aubinet, M., & Leuning, R. (2012). The eddy covariance method. In M. Aubinet, T. Vesala, & D. Papale (Eds.), Eddy covariance (pp. 1–19). Springer Netherlands. https://doi.org/10.1007/978-94-007-2351-1\_1
- Foken, T., Göckede, M., Mauder, M., Mahrt, L., Amiro, D., & Munger, J. W. (2004). Post-field data quality control. In Handbook of micrometeorology: A guide for surface flux measurement and analysis (pp. 181–208).
- Foken, Th., & Wichura, B. (1996). Tools for quality assessment of surface-based flux measurements. Agricultural and Forest Meteorology, 78(1-2), 83-105. https://doi.org/10.1016/0168-1923(95)02248-1
- Fujisaki-Manome, A., Mann, G. E., Anderson, E. J., Chu, P. Y., Fitzpatrick, L. E., Benjamin, S. G., et al. (2020). Improvements to lake-effect snow forecasts using a one-way air–lake model coupling approach. *Journal of Hydrometeorology*, 21(12), 2813–2828. https://doi.org/10.1175/ JHM-D-20-0079.1
- Garratt, J. (1994). Review: The atmospheric boundary layer. Earth-Science Reviews, 37(1-2), 89–134. https://doi.org/10.1016/0012-8252(94) 90026-4
- Garratt, J. R. (1977). Review of drag coefficients over oceans and continents. *Monthly Weather Review*, 105(7), 915–929. https://doi.org/10.1175/1520-0493(1977)105<0915:RODCOO>2.0.CO;2
- Geernaert, G. L. (1990). Bulk parameterizations for the wind stress and heat fluxes. In G. L. Geernaert & W. L. Plant (Eds.), *Surface waves and fluxes* (pp. 91–172). Springer Netherlands. https://doi.org/10.1007/978-94-009-2069-9\_5
- Glazunov, A. V., & Stepanenko, V. M. (2015). Large-eddy simulation of stratified turbulent flows over heterogeneous landscapes. Izvestiya -Atmospheric and Oceanic Physics, 51(4), 351–361. https://doi.org/10.1134/S0001433815040027
- Godfrey, J. S., & Beljaars, A. C. M. (1991). On the turbulent fluxes of buoyancy, heat and moisture at the air-sea interface at low wind speeds. Journal of Geophysical Research, 96(C12), 22043. https://doi.org/10.1029/91JC02015
- Golub, M., Desai, A. R., Vesala, T., Mammarella, I., Ojala, A., Bohrer, G., et al. (2021). New insights into diel to interannual variation in carbon dioxide emissions from lakes and reservoirs [Preprint]. Environmental Sciences. https://doi.org/10.1002/essoar.10507313.1
- Grachev, A. A., Bariteau, L., Fairall, C. W., Hare, J. E., Helmig, D., Hueber, J., & Lang, E. K. (2011). Turbulent fluxes and transfer of trace gases from ship-based measurements during TexAQS 2006. *Journal of Geophysical Research*, 116(D13), D13110. https://doi.org/10.1029/2010JD015502
- Grachev, A. A., Fairall, C. W., & Larsen, S. E. (1998). On the determination of the neutral drag coefficient in the convective boundary layer. Boundary-Layer Meteorology, 86(2), 257–278. https://doi.org/10.1023/A:1000617300732
- Graf, W. H., Merzi, N., & Perrinjaquet, C. (1984). Aerodynamic drag: Measured at a nearshore platform on Lake of Geneva. Archives for Meteorology, Geophysics, and Bioclimatology Series A, 33(2–3), 151–173. https://doi.org/10.1007/BF02257722
- Grant, L., Vanderkelen, I., Gudmundsson, L., Tan, Z., Perroud, M., Stepanenko, V. M., et al. (2021). Attribution of global lake systems change to anthropogenic forcing. *Nature Geoscience*, 14(11), 849–854. https://doi.org/10.1038/s41561-021-00833-x
- Guseva, S., Armani, F., Desai, A. R., Dias, N. L., Friborg, T., Iwata, H., et al. (2022). Data associated with article 'Bulk Transfer Coefficients Estimated from Eddy-Covariance Measurements Over Lakes and Reservoirs' by Guseva et al., 2022 (Version V2) [Dataset]. Zenodo. https:// doi.org/10.5281/ZENODO.6597828
- Guseva, S., Casper, P., Sachs, T., Spank, U., & Lorke, A. (2021). Energy flux paths in lakes and reservoirs. Water, 13(22), 3270. https://doi. org/10.3390/w13223270
- Harbeck, G. E. (1962). A practical field technique for measuring reservoir evaporation utilizing mass-transfer theory. 101-105.
- Heikinheimo, M., Kangas, M., Tourula, T., Venäläinen, A., & Tattari, S. (1999). Momentum and heat fluxes over lakes Tämnaren and Råksjö determined by the bulk-aerodynamic and eddy-correlation methods. *Agricultural and Forest Meteorology*, 98(99), 521–534. https://doi. org/10.1016/S0168-1923(99)00121-5
- Hicks, B. B. (1972). Some evaluations of drag and bulk transfer coefficients over water bodies of different sizes. *Boundary-Layer Meteorology*, 3(2), 201–213. https://doi.org/10.1007/BF02033919
- Högström, U. (1988). Non-dimensional wind and temperature profiles in the atmospheric surface layer: A re-evaluation. In B. B. Hicks (Ed.), Topics in micrometeorology. A Festschrift for arch dyer (pp. 55–78). Springer Netherlands. https://doi.org/10.1007/978-94-009-2935-7\_6
- Hwang, P. A. (2004). Influence of wavelength on the parameterization of drag coefficient and surface roughness. *Journal of Oceanography*, 60(5), 835–841. https://doi.org/10.1007/s10872-005-5776-3
- Jöhnk, K. D., Huisman, J., Sharples, J., Sommeijer, B., Visser, P. M., & Stroom, J. M. (2008). Summer heatwaves promote blooms of harmful cyanobacteria: Heatwaves promote harmful cyanobacteria. Global Change Biology, 14(3), 495–512. https://doi.org/10.1111/j.1365-2486.2007.01510.x
- Kaimal, J. C., & Finnigan, J. J. (1994). Atmospheric boundary layer flows: Their structure and measurement. Oxford University Press. https:// doi.org/10.1093/oso/9780195062397.001.0001
- Kantha, L., & Clayson, C. (2000). Small scale processes in geophysical Fluid flows (1st ed., Vol. 67). Academic Press.
- Kato, H., & Takahashi, H. (1981). Local climate near the small lake. Journal of Agricultural Meteorology, 37(1), 29–37. https://doi.org/10.2480/ agrmet.37.29
- Kuznetsova, A., Baydakov, G., Papko, V., Kandaurov, A., Vdovin, M., Sergeev, D., & Troitskaya, Y. (2016). Field and numerical study of the windwave regime on the Gorky Reservoir. *Geography, Environment, Sustainability*, 9(2), 19–37. https://doi.org/10.15356/2071-9388\_02v09\_2016\_02
- Large, W. G., & Pond, S. (1981). Open ocean momentum flux measurements in moderate to strong winds. *Journal of Physical Oceanography*. 11(3), 324–336. https://doi.org/10.1175/1520-0485(1981)011<0324:OOMFMI>2.0.CO;2
- Lee, X., Liu, S., Xiao, W., Wang, W., Gao, Z., Cao, C., et al. (2014). The Taihu eddy flux network: An observational program on energy, water, and greenhouse gas fluxes of a large freshwater lake. *Bulletin of the American Meteorological Society*, 95(10), 1583–1594. https://doi. org/10.1175/BAMS-D-13-00136.1



- Lenschow, D. H., Mann, J., & Kristensen, L. (1994). How long is long enough when measuring fluxes and other turbulence statistics? *Journal of Atmospheric and Oceanic Technology*, *11*(3), 661–673. https://doi.org/10.1175/1520-0426(1994)011<0661:HLILEW>2.0.CO;2
- Li, Z., Lyu, S., Zhao, L., Wen, L., Ao, Y., & Wang, S. (2016). Turbulent transfer coefficient and roughness length in a high-altitude lake, Tibetan Plateau. *Theoretical and Applied Climatology*, 124(3–4), 723–735. https://doi.org/10.1007/s00704-015-1440-z
- LI-COR, Inc. (2021). EddyPro® software (version 7.0) [Software], LI-COR. Retrieved from https://www.licor.com/env/support/EddyPro/home.html Lin, W., Sanford, L. P., Suttles, S. E., & Valigura, R. (2002). Drag coefficients with fetch-limited wind waves. *Journal of Physical Oceanography*, 32(11), 3058–3074. https://doi.org/10.1175/1520-0485(2002)032<3058:DCWFLW>2.0.CO;2
- Liu, C., Li, Y., Gao, Z., Zhang, H., Wu, T., Lu, Y., & Zhang, X. (2020). Improvement of drag coefficient calculation under near-neutral conditions in light winds over land. *Journal of Geophysical Research: Atmospheres*, 125(24). https://doi.org/10.1029/2020JD033472
- Ljungemyr, P., Gustafsson, N., & Omstedt, A. (1996). Parameterization of lake thermodynamics in a high-resolution weather forecasting model. *Tellus A*, 48(5), 608–621. https://doi.org/10.1034/j.1600-0870.1996.t01-4-00002.x
- Lükő, G., Torma, P., Krámer, T., Weidinger, T., Vecenaj, Z., & Grisogono, B. (2020). Observation of wave-driven air-water turbulent momentum exchange in a large but fetch-limited shallow lake. Advances in Science and Research, 17, 175–182. https://doi.org/10.5194/asr-17-175-2020 MacKay, M. D. (2019). Incorporating wind sheltering and sediment heat flux into 1-D models of small boreal lakes: A case study with the Cana-
- dian Small Lake Model V2.0. *Geoscientific Model Development*, *12*(7), 3045–3054. https://doi.org/10.5194/gmd-12-3045-2019 Mammarella, I., Nordbo, A., Rannik, Ü., Haapanala, S., Levula, J., Laakso, H., et al. (2015). Carbon dioxide and energy fluxes over a small boreal
- Mammarella, I., Nordbo, A., Rannik, U., Haapanala, S., Levula, J., Laakso, H., et al. (2015). Carbon dioxide and energy fluxes over a small boreal lake in Southern Finland: CO<sub>2</sub> and energy fluxes over lake. *Journal of Geophysical Research: Biogeosciences, 120*(7), 1296–1314. https://doi. org/10.1002/2014JG002873
- Mammarella, I., Peltola, O., Nordbo, A., Järvi, L., & Rannik, Ü. (2016). Quantifying the uncertainty of eddy covariance fluxes due to the use of different software packages and combinations of processing steps in two contrasting ecosystems. *Atmospheric Measurement Techniques*, 9(10), 4915–4933. https://doi.org/10.5194/amt-9-4915-2016
- Mauder, M., & Foken, T. (2015a). Eddy-covariance software TK3 [Software]. Zenodo. https://doi.org/10.5281/ZENODO.20349
- Mironov, D., Heise, E., Kourzeneva, E., Ritter, B., Schneider, N., & Terzhevik, A. (2010). Implementation of the lake parameterization scheme FLake into the numerical weather prediction model COSMO. *Boreal Environment Research*, 15, 218–230.
- Molod, A., Salmun, H., & Waugh, D. W. (2003). A new look at modeling surface heterogeneity: Extending its influence in the vertical. Journal of Hydrometeorology, 4(5), 810–825. https://doi.org/10.1175/1525-7541(2003)004<0810:ANLAMS>2.0.CO;2
- Monin, A. S., & Obukhov, A. M. (1954). Basic laws of turbulent mixing in the surface layer of the atmosphere. Contrib. Geophys. Inst. Acad. Sci. USSR, 24, 163–187.
- Nordbo, A., Launiainen, S., Mammarella, I., Leppäranta, M., Huotari, J., Ojala, A., & Vesala, T. (2011). Long-term energy flux measurements and energy balance over a small boreal lake using eddy covariance technique. *Journal of Geophysical Research*, 116(D2), D02119. https:// doi.org/10.1029/2010JD014542
- Obukhov, A. M. (1971). Turbulence in an atmosphere with a non-uniform temperature. *Boundary-Layer Meteorology*, 2(1), 7–29. https://doi.org/10.1007/BF00718085
- Panin, G. N., Nasonov, A. E., & Foken, T. (2006). Evaporation and heat exchange of a body of water with the atmosphere in a shallow zone. Izvestiya - Atmospheric and Oceanic Physics, 42(3), 337–352. https://doi.org/10.1134/S0001433806030078
- Paulson, C. A. (1970). The mathematical representation of wind speed and temperature profiles in the unstable atmospheric surface layer. Journal of Applied Meteorology, 9(6), 857–861. https://doi.org/10.1175/1520-0450(1970)009<0857:TMROWS>2.0.CO;2
- Pawlowicz, R. (2020). M\_Map: A mapping package for MATLAB (1.4m) [Software]. MATLAB. Available online at www.eoas.ubc.ca/~rich/ map.html
- Read, J. S., Hamilton, D. P., Desai, A. R., Rose, K. C., MacIntyre, S., Lenters, J. D., et al. (2012). Lake-size dependency of wind shear and convection as controls on gas exchange. *Geophysical Research Letters*, 39(9). https://doi.org/10.1029/2012GL051886
- Read, J. S., Winslow, L. A., Hansen, G. J. A., Van Den Hoek, J., Hanson, P. C., Bruce, L. C., & Markfort, C. D. (2014). Simulating 2368 temperate lakes reveals weak coherence in stratification phenology. *Ecological Modelling*, 291, 142–150. https://doi.org/10.1016/j.ecolmodel.2014.07.029
- Rosentreter, J. A., Borges, A. V., Deemer, B. R., Holgerson, M. A., Liu, S., Song, C., et al. (2021). Half of global methane emissions come from highly variable aquatic ecosystem sources. *Nature Geoscience*, 14(4), 225–230. https://doi.org/10.1038/s41561-021-00715-2
- Sahlée, E., Rutgersson, A., Podgrajsek, E., & Bergström, H. (2014). Influence from surrounding land on the turbulence measurements above a lake. Boundary-Layer Meteorology, 150(2), 235–258. https://doi.org/10.1007/s10546-013-9868-0
- Salgado, R., & Le Mogne, P. (2010). Coupling the FLake model to the surfax externalized surface model. *Boreal Environment Research*, 15, 231–244.
- Schlichting, H. (1968). Boundary-layer theory (6th ed.). McGraw-Hill Book Company.
- Simon, A. (1997). Turbulent mixing in the surface boundary layer of lakes [Dissertation No 12272]. Swiss Federal Institute of Technology.
- Smith, S. D., Fairall, C. W., Geernaert, G. L., & Hasse, L. (1996). Air-sea fluxes: 25 years of progress. Boundary-Layer Meteorology, 78(3–4), 247–290. https://doi.org/10.1007/BF00120938
- Spank, U., Hehn, M., Keller, P., Koschorreck, M., & Bernhofer, C. (2020). A Season of eddy-covariance fluxes above an extensive water body based on observations from a floating platform. *Boundary-Layer Meteorology*, 174(3), 433–464. https://doi.org/10.1007/s10546-019-00490-z
- Stepanenko, V., Jöhnk, K. D., Machulskaya, E., Perroud, M., Subin, Z., Nordbo, A., et al. (2014). Simulation of surface energy fluxes and stratification of a small boreal lake by a set of one-dimensional models. *Tellus A: Dynamic Meteorology and Oceanography*, 66(1), 21389. https:// doi.org/10.3402/tellusa.v66.21389
- Strub, T. P., & Powell, T. M. (1987). The exchange coefficients for latent and sensible heat flux over lakes: Dependence upon atmospheric stability. Boundary-Layer Meteorology, 40(4), 349–362. https://doi.org/10.1007/BF00116102
- Stull, R. B. (Ed.) (1988). An introduction to boundary layer Meteorology. Springer Netherlands. https://doi.org/10.1007/978-94-009-3027-8
- Sun, J., Lenschow, D. H., Mahrt, L., Crawford, T. L., Davis, K. J., Oncley, S. P., et al. (1997). Lake-induced atmospheric circulations during BOREAS. Journal of Geophysical Research, 102(D24), 29155–29166. https://doi.org/10.1029/97JD01114
- Sun, X.-M., Zhu, Z.-L., Wen, X.-F., Yuan, G.-F., & Yu, G.-R. (2006). The impact of averaging period on eddy fluxes observed at ChinaFLUX sites. Agricultural and Forest Meteorology, 137(3–4), 188–193. https://doi.org/10.1016/j.agrformet.2006.02.012
- Thiery, W., Davin, E. L., Seneviratne, S. I., Bedka, K., Lhermitte, S., & van Lipzig, N. P. M. (2016). Hazardous thunderstorm intensification over Lake Victoria. Nature Communications, 7(1), 12786. https://doi.org/10.1038/ncomms12786
- Vesala, T., Huotari, J., Rannik, Ü., Suni, T., Smolander, S., Sogachev, A., et al. (2006). Eddy covariance measurements of carbon exchange and latent and sensible heat fluxes over a boreal lake for a full open-water period. *Journal of Geophysical Research*, 111(D11), D11101. https:// doi.org/10.1029/2005JD006365
- Vickers, D., & Mahrt, L. (1997). Fetch limited drag coefficients. *Boundary-Layer Meteorology*, 85(1), 53-79. https://doi.org/10.1023/A:1000472623187



- von Salzen, K., Claussen, M. M., & Schlünzen, K. H. (1996). Anwendung des Konzepts der Blendhöhe zur Berechnung von bodennahen Flüssen in einem mesoskaligen Modell. *Meteorologische Zeitschrift*, 5(2), 60–66. https://doi.org/10.1127/metz/5/1996/60
- Wei, Z., Miyano, A., & Sugita, M. (2016). Drag and bulk transfer coefficients over water surfaces in light winds. Boundary-Layer Meteorology, 160(2), 319–346. https://doi.org/10.1007/s10546-016-0147-8
- Woolway, R. I., & Merchant, C. J. (2019). Worldwide alteration of lake mixing regimes in response to climate change. Nature Geoscience, 12(4), 271–276. https://doi.org/10.1038/s41561-019-0322-x
- Woolway, R. I., Verburg, P., Lenters, J. D., Merchant, C. J., Hamilton, D. P., Brookes, J., et al. (2018). Geographic and temporal variations in turbulent heat loss from lakes: A global analysis across 45 lakes. *Limnology & Oceanography*, 63(6), 2436–2449. https://doi.org/10.1002/ lno.10950
- Woolway, R. I., Verburg, P., Merchant, C. J., Lenters, J. D., Hamilton, D. P., Brookes, J., et al. (2017). Latitude and lake size are important predictors of over-lake atmospheric stability: Atmospheric stability above lakes. *Geophysical Research Letters*, 44(17), 8875–8883. https:// doi.org/10.1002/2017GL073941
- Wu, J. (1988). Wind-stress coefficients at light winds. Journal of Atmospheric and Oceanic Technology, 5(6), 885–888. https://doi.org/10.1175/ 1520-0426(1988)005<0885:WSCALW>2.0.CO;2
- Wu, J. (1994). The sea surface is aerodynamically rough even under light winds. Boundary-Layer Meteorology, 69(1–2), 149–158. https://doi. org/10.1007/BF00713300
- Wu, L., Breivik, Ø., & Rutgersson, A. (2019). Ocean-wave-atmosphere interaction processes in a fully coupled modeling system. Journal of Advances in Modeling Earth Systems, 11(11), 3852–3874. https://doi.org/10.1029/2019MS001761
- Wüest, A., & Lorke, A. (2003). Small-scale hydrodynamics in lakes. Annual Review of Fluid Mechanics, 35(1), 373–412. https://doi.org/10.1146/ annurev.fluid.35.101101.161220
- Xiao, W., Liu, S., Wang, W., Yang, D., Xu, J., Cao, C., et al. (2013). Transfer coefficients of momentum, heat and water vapour in the atmospheric surface layer of a large freshwater lake. *Boundary-Layer Meteorology*, 148(3), 479–494. https://doi.org/10.1007/s10546-013-9827-9
- Zhu, P., & Furst, J. (2013). On the parameterization of surface momentum transport via drag coefficient in low-wind conditions. *Geophysical Research Letters*, 40(11), 2824–2828. https://doi.org/10.1002/grl.50518
- Zilitinkevich, S., & Calanca, P. (2010). An extended similarity theory for the stably stratified atmospheric surface layer. Quarterly Journal of the Royal Meteorological Society, 126(566), 1913–1923. https://doi.org/10.1002/qj.49712656617